Author’s response to reviews

Title: Remarkable increase of musculoskeletal disorders among soldiers preparing for international missions – Comparison between 2002 and 2012

Authors:

Alexandra Halvarsson (alexandra.halvarsson@ki.se)
Monika Seth (monika_seth@hotmail.com)
Matthias Tegern (matthias.tegern@ki.se)
Lisbet Broman (lisbet.broman@ki.se)
Larsson Helena (helena.larsson.2@ki.se)

Version: 2 Date: 26 Sep 2019

Author’s response to reviews:

Dear editor,

We have revised the manuscript in accordance to the two comments we received, i.e. we have added that all authors have read and approved the final version of the manuscript and added text in the method section about control of normal distributed data.

Kind regards,

Alexandra Halvarsson

On behalf of all the authors